meta-data

opened file_no open

. NIL

rel_name cardinality width file_size free_slot REL_DESC open_count and rel_desc in OPEN_FILE_DESC record

record.c lseek, read, write

calls.)

(1) get_ _ _

(2) put_ _ _

(3) append_ _ _no)

file_size free_slot appropriately. (4) delete_ _ _no)

(5) get_next_ _ _rec_ _rec_no) last_rec_no. Get_next_record passes OKAY ERR_EOF.

(6) find_ _ _rec_ _rec_ _ _arg) last_rec_no for TRUE tion cond_fct TRUE FALSE. Find_record cond_arg NULL find_record Find_record OKAY this_rec_no *record ERR_EOF.

file open, close unlink UNIX file.c relation rel_name open_count RES_FILE_NO and MAX_FILE_NO (see defs.h

(1) find_open_ _no) get_record() and put_record() last_seek_pos lseek(),

(2) create_ _no)

(3) open_ _no)

(4) close_ _no) (5) destroy_

attribute catalog rel_cat_file_no and attr_cat_file_no. arity T_INT, T_REAL T_CHAR defs.h arity-1

schema and relation database

The schema relation and database

relation routine schema routines).

(1) save_ _ _ _desc)

(2) append_ _no)

alloc_record() and dealloc_record() append_record().) (3) find_ _no)

find_record alloc_schema(), alloc_ _record().

(4) put_ _no) put_record

(5) destroy_ _no) delete_record. relation.c:

(1) open_ _no) open_count

(2) close_ _no) (3) create_ argc argv points

WORD_SIZE defs.h my_table left right, create_relation create_ _ _relation create_relation1 typed_def (in

(4) destroy_ db_path modify file.c (1) create_

(2) destroy_ (3) open_ (4) close_

append.c, print.c, help.c, load.c, dump.c, delete.c (1) append_

atoi() and atof()

(2) type_min_print_width and typed_print rou-

(3) argc = argc

(4) argv [argc-1] create_relation Load typed_scan argc argv[0]

(5) . Dump uses typed_print dump load. (6) select (7)

argv update() update.c.

destroy_relation.

(1) argv [3i+4] (2) argv [3], (3) and and .

(4)

in _INT." (5) , is unique qsort()

The buffer least-recently-used

fixed

releases the

dirty

Get_record and put_record

get_record and put_record Append_record Delete_record get_record and put_record. Get_next_record and find_record buffer. bcopy bu_io lseek, read, write.

close_file buffer.c

(1) bu_ _ _ _no) frame_no io (2) _ _ _no) -1 (3) _ _ _no)

Bu_alloc 1 (4) bu_

(5) bu_ _ _

invoking bu_alloc addr append_record bu_ .

(6) bu_ _ _ addr bu_read invokes bu_alloc and bu_io bu_ .

(7) bu_ _ _no)

(8) bu_ _ _no)

(9) bu_ _no) Hints defs.h

run. test1.c

dbmgr syntax.

get_record and put_record io_record()

find_record test1.c.

baseof

misc.c

Make Makefile -DDEBUG -Ddebug_record _record. RCS or make

follows.

cb indent lint

defs.h defs.h globals.c close lseekor read

OKAYor ERR_EOF

defs.h my_defs.h

_

.

/n/homeserver/w/cs425/1996/dbmgr.tar.Z

_425 ls studentversions_425

docs and src docs src ls The ls

ls

test1.c

makefile

*.o .